Author's response to reviews

Title: Clinical apparent thrombosis as a complication of central venous access in pediatric patients with malignancies: A 5-year single-center experience

Authors:

Verena Wiegering (Wiegering_V@ukw.de)
Sophie Schmid (sophie_schmid@web.de)
Oliver Andres (andres_O@ukw.de)
clemens Wirth (Wirth_C2@ukw.de)
Armin Wiegering (Wiegering_a@ukw.de)
Thomas Meyer (Meyer_T@ukw.de)
Beate Winkler (Winkler_B@ukw.de)
Paul G Schlegel (Schlegel_P@ukw.de)
Matthias Eyrich (Eyrich_M@ukw.de)

Version:2 Date:31 March 2014

Author's response to reviews:

We changed as quoted the title page, the line numbering and the heading for the conclusion. We hope it is now in according to yur guidelines. Please do not hesitate to contact us if you have any questions.